

Sveriges lantbruksuniversitet Swedish University of Agricultural Sciences

**Department of Economics** 

The economic value of the effect of weather variability on forest ecosystem services in Sweden

Jean Damascene Uwizeye



Master's thesis • 30 credits Agricultural programme – Economics and Management Degree project/SLU, Department of Economics, 1265 • ISSN 1401-4084 Uppsala, Sweden 2019

## The economic value of the effect of weather variability on forest ecosystem services in Sweden

Jean Damascene Uwizeye

| Supervisor: | Franklin Amuakwa Mensah,<br>Swedish University of Agricultural Sciences,<br>Department of Economics |
|-------------|---|
| Examiner:   | Jens Rommel, Swedish University of Agricultural Sciences,   |

Department of Economics

Credits: 30 hec Level: A2E Course title: Master thesis in Economics – Environmental Programme, Economics and Management Course code: EX0907 Programme/Education: Agricultural programme – Economics and Management Faculty: Faculty of Natural Resources and Agricultural Sciences

Place of publication: Uppsala Year of publication: 2019 Name of Series: Master's degree/SLU, Department of Economics No: 1265 ISSN 1401-4084 Online publication: http://stud.epsilon.slu.se

**Key words:** carbon sequestration, economics, forest growth, management practices, Sweden, timber, weather conditions

## Abstract

In Sweden, production forests primarily consist of planted coniferous trees often mixed with naturally regenerated broadleaves. The predominating boreal conifer species are Norway spruce, and Scots pine which constitutes 40.8% and 39% of the total productive forests cover respectively. This thesis aims to estimate the effect of weather variability and management practices on forest productivity and to calculate the total economic value of forest ecosystem services of timber and carbon sequestration in Sweden. The analysis is carried out in two steps. Firstly, to estimate the impact of weather conditions and management treatments on forest growth using a theoretically sound linearized logistic growth model. Secondly, to quantify the total economic value of the timber and carbon sequestration based on their unit net prices using the static additive model approach.

The findings suggest that forest productivity in the boreal region increases with the rise of temperature and the length of the vegetation period. Also, the management system of scarification contributes to the increase in standing volume per hectare. The long-term effect of scarification can last from the beginning of young planted seedlings until they reach high levels of productivity. Accordingly, the increase of weather variability of temperature and the length of the vegetation period on forest productivity contribute to the total economic benefit mainly through the incremental value of carbon sequestration. The total economic value in the base scenario is reported to be SEK 1065/m<sup>3</sup>/ha. In average, the total value increases by about 11% with a one-unit increase of temperature and around 1.5% with a one-unit increase of the length of the vegetation period relative to the base scenario. However, the value of timber remains constant and equals to SEK 923/m<sup>3</sup>/ha in all calculation scenarios.

In 2018, the economic value of timber production in the entire country constituted 23% of the gross domestic product (GDP). Because of the nature of local weather conditions which vary across different regions in the country, this study could not easily estimate the overall total economic value of both timber and carbon sequestration at a national level. The results of this study provide incentives for future research to use mathematical relationships that could not only capture potential economic benefits but also would optimize management decisions which produce long-term impact on sustainable forest ecosystems.

## Abbreviations

CO2e – Carbon dioxide equivalents. FMOLS- Fully Modified Ordinary Least Square GHG- Greenhouse gases Ibid – ibidem (in the same thesis). mm- Millimeter NPP- Net Primary Production SMHI- Swedish Meteorological and Hydrological Institute

# Table of Contents

| ABSTRACT  |   | . 111                                |
|---|---|--------------------------------------|
| ABBREVIATIO   | NS  | .IV                                  |
| 1 INTRODU   |   | 1                                    |
| 1.1<br>1.2<br>1.3   | Problem background<br>Problem statement<br>The objective of the study   | 1<br>2<br>2                          |
| 1.4   | Disposition   | 3                                    |
| 2 OVERVIE<br>SWEDEN   | W OF FOREST MANAGEMENT PRACTICES AND WEATHER CONDITIONS IN  | 4                                    |
| 2.1<br>2.2<br>2.3<br>2.3.1<br>2.3.2<br>2.3.3<br>2.4<br>2.4.1<br>2.4.2<br>2.4.3<br>2.4.4 | Forestry Industry in Sweden<br>Forest Ecosystem Services<br>Forest Management Practices<br>Fertilization<br>Thinning<br>Scarification<br>Weather variability<br>Temperature<br>Precipitation<br>The length of the vegetation period<br>Windstorms | 4<br>5<br>6<br>7<br>8<br>8<br>9<br>9 |
| 3 LITERATI  | JRE REVIEW  | 11                                   |
| 3.1<br>3.2<br>3.3   | Management practices and forest growth<br>Weather patterns and forest growth<br>Carbon Sequestration  | 11<br>14<br>15                       |
| 4 METHOD  | OLOGY AND EMPIRICAL DATA  | 16                                   |
| 4.1<br>4.2  | Methodology<br>Empirical data   | 16<br>19                             |
| 5 RESULTS   | ANALYSIS  | 23                                   |
| 5.1<br>5.2<br>5.3   | Management practices and forest productivity<br>Weather variability and forest productivity<br>Estimation of the total economic value   | 25<br>26<br>27                       |
| 6 DISCUSS   | ION   | 30                                   |
| 7 CONCLUS   | SIONS   | 33                                   |
| REFERENCES  |   | 34                                   |
| ACKNOWLED   | GMENT   | 41                                   |
| APPENDIX  |   | 42                                   |

# List of figures

| Figure 1. The process of calculating the total economic value | 3 |
|---|---|
| Figure 2. Structure of the thesis                             | 3 |

## List of tables

| Table 1. Descriptive statistics   | 18 |
|---|----|
| Table 2. Regression results from fully modified ordinary least square (FMOLS) | 24 |
| Table 3: Calculated total economic value                                      |    |

## 1 Introduction

The forestry has long been an important sector that offers various benefits to society. In recent years, dramatic changes in the forestry industry have vastly increased because of its multiple uses such as commercial value, maintaining national biodiversity, importance in water catchments, and carbon sequestration. These changes are partly due to significant impacts of climate variability on forest growth and its compositions (Coops *et al.* 1998). Thus, the management of forest resources should seek the balance of yields from as many benefits as possible. Furthermore, we should move beyond maximizing the merchantable biomass and find answers to several optimization questions regarding the benefits provided by forest ecosystems. To understand the total economic value, we need to consider the expansion of these paramount ecosystem services.

## 1.1 Problem background

Swedish forests are even-aged monoculture with the primary dominating boreal conifer species of Norway spruce (Picea abies(L.) Karst.) and Scots pine(Pinus sylvestric L.). The principal forest management practices are fertilization, scarification, and thinning in the entire country. Furthermore, total harvesting of the entire stands and re-planting new seedlings are massive operations of intensive management systems. On the other hand, weather variability is an essential factor that affects forests in different ways. For example, a rising temperature could affect the forest in two ways: by changing forest productivity of the actual forest biomass or by changing forest distributions in different geographical areas (Eriksson & Vesterberg 2016). A slow but continuous rise in temperature could also have different impacts on forest site conditions resulting in the increasing of forest productivity which is then, at a certain threshold, followed by a loss of carbon storage due to an elevated temperature (Daly *et al.* 2000).

The forest cover is more than half of the country's total land surface, meaning that the forestry sector is of a great contributor to the national economy. For instance, Sweden is the 5<sup>th</sup> highest combined exporter of pulp, paper, and forest products, after the United States, China, Germany, and Canada. The Swedish forest industry represents about 10 percent of exports, revenue, employment, and value-added of the entire economy (WITS 2017).

In their study of managing forest in uncertain times, Bellassen & Luyssaert (2014) pointed out that there is a shortage of knowledge of the impacts of temperature similar to any other weather patterns such as precipitation on tree growth. They found that changing forest management practices is of great importance only at the local level (national and regional) not at the global level. The present study aims to estimate the total economic value of forest ecosystem services from the effects of weather patterns and management practices on forest productivity in Sweden for about 50 years.

### 1.2 Problem statement

The long-term forest management decisions that seek to maximize economic value should consider weather variability as an essential factor in the dynamics of forest growth. Forest trees in boreal zones are growing below the optimum weather conditions, and therefore, they react with a higher growth rate to an increase in temperature (Way & Oren 2010). However, under extreme weather conditions, there is a high risk that they will respond with a substantial fall in the growth rate (Scholze *et al.* 2006). These dynamics of forest productivity due to weather patterns and management practices have consequences of variations in the economic development of the sector. This investigation of the effect of weather patterns and silvicultural systems on forest growth is indeed intended to take the current scientific knowledge to another level. The motive is to understand their contribution in changing the forest growth rate and the overall magnitude on the entire ecosystems, which in combination will serve for quantifying the total economic value of forest ecosystem services.

### 1.3 The objective of the study

The main objectives of this study are, therefore, to (i) estimate the effects of forest management practices and weather patterns on forest growth. Furthermore, (ii) quantify the total economic value of timber and carbon sequestration provided by forest ecosystems in Sweden (see fig.1).



Figure 1. The process of calculating the total economic value.

The panel data on forests, weather patterns and intensive management practices in four regions of Sweden; norra Norrland, södra Norrland, Svealand and Götaland, (see figure A1) were used, starting from 1965 until 2013, to assess the effects of weather variability and silvicultural practices on forest growth using econometric modelling. It is then followed by quantifying the total economic value of timber and carbon storage using their respective unit process with the additivity approach.

## 1.4 Disposition

The rest of the work is organized in the following way to achieve the objectives of the thesis. Section 2 presents an overview of forest management practices and weather conditions. Section 3 reviews the literature. Section 4 illustrates the methodology and empirical data of the study, and section 5 describes analysis of the results, which is then followed by section 6 that presents discussions. Section 7 presents conclusions (see fig. 2)





## 2 Overview of forest management practices and weather conditions in Sweden

### 2.1 Forestry Industry in Sweden

The forest industry is one of the critical pillars of Sweden's business sectors and provides tens of thousands of jobs throughout the country. More than half of Sweden's total land area is the forest, with approximately 23.50 million hectares of productive forest land, and 4.62 million hectares of unproductive forest land. The most dominating tree species are spruce and pine each constituting 40.8 % and 39% of the total forest land, respectively. Forest resources productivity has doubled over the last 100 years, while forest harvest is about one percent of what is produced every year (Nilsson *et al.* 2018). Furthermore, the forest industry depends heavily on export market orientation with possible variations between 9-12% of the exports, employment, turnover and value addition compared to the relatively small share of import of forest resources and use of raw materials in the entire industry in Sweden. Also, this industry accounts for 17% of the total manufacturing sector (Nolander, 2018).

### 2.2 Forest Ecosystem Services

Sweden has intensive forest ecosystems resulting from its special weather conditions. To some extent, this is because the country experiences yearly winter frozen conditions and that there are relatively few tree species, unlike the diversity of corresponding forest ecosystems around the world. The habitats in the boreal forest follow a random walk of temporal changes of various environmental conditions, caused for example, by forest fires, and storms. However, these natural disturbances have experienced regular control over the last decades because of the well-organized development of the forestry industry (SFA 2015).

The forests play an essential role in binding carbon emissions that have gone into the atmosphere. While global forests suck up 30% of anthropogenic emissions every year, the Swedish forests offset more than 80% of the total country's emissions (Nolander 2018; *LULUCF* 2018). Carbon sequestration is an excellent and most prominent regulatory forest ecosystem service. Usually, it takes the value of either taxpayer-subsidies related carbon emissions or the social costs associated with greenhouse gases emissions (Brainard *et al.* 2009).

Besides that, forest carbon flow is a very complex mechanism of forest ecosystems. Some drivers of this flow process are, for example, weather variability, tree species, geographical location, and silvicultural treatments (Bellassen & Luyssaert 2014).

Also, carbon sequestration is the most critical ecosystem service provided by forests at least in recent years. The geographical location of forest plantations does not matter since the amount of carbon storage benefits at large scale around the globe. The annual storage of carbon emissions, mainly carbon dioxide, determines the fundamental value of this ecosystem service associated with the volume of growing tree biomass and the soil nutrients (Nolander and Lundmark, 2018; Brainard *et al.* 2009). It is worth noting that various tree species have different rates of growth, and accordingly, the amount of carbon sequestered is expected to vary with species. Additionally, the warmer the temperatures, usually the better conditions for trees to grow and to increase carbon storage.

### 2.3 Forest Management Practices

Forest management practices are mostly performed with the purpose to increase the productivity of tree biomass since these management activities lead to a higher net primary production of the stem-wood. Similarly, these practices reduce the level of heterotrophic respiration because parts of biomass removals perish and decay. As a result, it turns out that the effects of silvicultural treatments such as thinning and harvesting contribute to regulating carbon emissions (Luyssaert *et al.* 2007). It shows that forest site quality changes over time with the choice of management treatments and have significant effects on forest growth and therefore changing profitability.

The current Swedish forest policy constitutes two equally important main objectives linked to the legal request for intensive management practices. The first objective focuses on the increase of timber production, and the second objective equally well supports biodiversity conservation (SFA 2015). Several studies have questioned the optimal management practices that would drive towards maximization of carbon storage because the optimal rotation ages are estimated to be longer when they are associated with carbon sequestration, and they are shorter for optimal harvesting decision. Hence, the prevailing hypothesis that bioenergy is a net-zero carbon footprint is challenged (Bjørnstad & Skonhoft; McDermott *et al.* 2015).

#### 2.3.1 Fertilization

In Sweden, fertilization in forest sector was introduced in the late 1960s after extensive and thorough studies connecting various aspects of the boreal forests including physiology, yield, ecology, and nutrient (Tamm, C.-O, 1991; Enander KG, 2007). However, forest fertilization did not develop intensively as an operational management practice until the 1970s when fertilizers were used in more than 200 thousand hectares and then started dropping in the 1990s (Lindkvist *et al.* 2011). The analysis carried out for Holmen Skog AB indicated that while the selection of tree species and improved planting materials have substantially increased profitability with reasonable outlay in costs, fertilization proved to be much more profitable.

Using forest fertilizers to increase tree biomass of Norway spruce was proved to be nonbeneficial since it generates high cost at the beginning of the project (Eriksson *et al.* 2008). Therefore, the Swedish forest sector almost abandoned fertilization as a management practice. This is not only because it incurred higher investment cost but also because of its detrimental impacts on the environment. Despite the high costs and adverse effects on the environment, fertilization has regained popularity among Swedish forest owners in the past decade. Since 2010, statistics reports show that its consumption in the forest is at an all-time high (Nilsson *et al.* 2017). The current political agenda endorses the use of forest products. It also promotes non-fossil fuel (biofuels). These are incentives to boost the forest sector, which is one of the driving industries of the national economy, then also challenging global warming at a massive scale.

#### 2.3.2 Thinning

Thinning as forest management activity aims to enhance forest productivity, then boosting the profitability of the forest industry. This primary method has been used in Swedish forests over the last 200 years, most notably "thinning from below" (Wallentin 2007). Yet, several thinning studies have concluded that thinning methods also hurt total forest productivity. For example, there is a negative correlation between the growth of Norway spruce and Scots pine and the average amount of removals from a particular basal area (Makinen 2004; Mäkinen & Isomäki 2004; Nilsson 2010). Some exceptional of spruce species respond to light and moderate thinning by developing high tree volume productivity (Juodvalkis *et al.* 2005).

The thinning leads to a decrease of stand density which in return induces the remaining part of the tree to grow competitively relative to other available factors such as the nutrients, water and solar radiation (Savill 1997; Larson *et al.* 2001). Usually, we would not expect the thinning to raise the total productivity per unit area, but to some extent thinning would contribute to the development of high growth rate of trees and the yield of the log-sized timber (Makinen 2004; Mäkinen & Isomäki 2004). The beneficial effects of thinning depend on the amount and value of the removals, timing, and magnitude of precommercial thinning with additional costs of clear-cutting (Huuskonen & Hynynen 2006). In the forest sector, the intensity of thinning was improved during the 1960s, reflecting the mechanization of harvesting practices. In Finland for example, caring for younger forest stands is done at least in three successive thinning operations throughout the entire rotation period which can usually last up to 80 years (Tapio 2013)

#### 2.3.3 Scarification

Scarification has been the most popular and uninterrupted management method used in trees plantation in different parts of the country and has proved to be successful practice for various tree species (Nilsson *et al.* 2011, 2018). This mechanical management practice consists mainly of two different major categories: disc trenching and mounding. The former is suitable in all four different agro-ecological zones of this study, and the latter can be applied basically in an environment with enough moisture and moderately stony soils. These two method categories produce different quality of forest plantations. The differences are attributed to the particular quality of the forest site and other bounding factors linked to these practices (Örlander *et al.* 1990). While disc trenching generates insufficient nutrients and an excess of water compared to what plantations need, mounding could increase the probability of drought stress specifically in newly established plantations on highland spots (Nilsson & Örlander 1999).

Thus, applying these two silvicultural practices can have from short term to long term adverse effects due to a sharp decrease of nutrients as well as to a high degree of the site disturbances (Johansson 1994). Nevertheless, introducing soil inversion as another type of scarification reduces the adverse effects caused by the previous two methods.

The purpose is to level seedlings planting spot and the nearest soil ground so that the early planted seedlings will develop desirably (Nordborg *et al.* 2003; Johansson *et al.* 2006, 2013b)

### 2.4 Weather variability

#### 2.4.1 Temperature

It is necessary to consider forest planting season when selecting tree species to lessen potential effects of extreme events and where possible to counter damages altogether. Accordingly, for forest growth, the average values should be combined with short-term extreme weather variations in order to produce the factual data which are likely imminent to a study that can assess the relationship effect. An extraordinary drastic frost may put some trees in an abiotic state. In the same way, unexpected hot days might stop the physiological process of some tree species in the boreal such as the drought that can occur for a few weeks and kill several trees. Thus, as for tree growth is concerned, weather patterns need to include the exceptional events that have occurred to capture their effects on the actual value of forest growth.

Basically, during a period of modest warm weather condition, the boreal forest growth rate tends to increase, whereas the scorching sun has adverse effects on forest productivity (Eriksson & Vesterberg 2016). This study tests the same hypothesis by analysing the effects of the average temperatures on the growth rate of the forest for an extended period. The Scandinavian region has likely experienced global warming since an upward shift in temperature can be perceived. These patterns of weather illustrate evidence about the speed at which glaciers are retreating over time while temporal scorching days have been recorded repeatedly during the summers.

#### 2.4.2 Precipitation

On the other hand, the average amount of precipitation corresponds to the rain expected to fall on different occasions annually. Forests grow better in moist areas with easy access to enough water. Based on practical experience, forest ecologists came up with a rule of thumb that forests rarely grow in dry geographical locations where the average rainfall gets lower than about 700 millimeters (mm) annually.

Nonetheless, this may rely relatively upon the amount of rain that falls periodically associated with its level of evaporation as well as the readiness of soil to save this water since enough soil water storage implies essential effects in the growth of forests (Landsberg & Waring 2014).

In Sweden, the average amount of rainfall per year reaches between 500 and 800 mm, which characterizes a dry geo-climatic zone compared to the world average annual rainfall.

#### 2.4.3 The length of the vegetation period

The span of the vegetation period is also called as the growing season in the forest sector. According to the Swedish Meteorological and Hydrological Institute (SMHI), the length of vegetation period starts the first day in the year when the daily average temperature is at least +5 degrees for at least four days. This period ends the last day in the year when the daily average temperature is at least +5 degree for at least four days. The length of the vegetation period is the number of days from start to end, including the start and the end days.

Unfortunately, during a particular year, temperature variation can largely exceed the average temperature values causing difficulties to determine the precise point in time of which the average daily heat overreaches a homogeneity value. For instance, the average temperature from day to day over ten years has provided consistent temperature variability in southern Sweden, more specifically in Götaland and southern Svealand. Unlike the extent to which length of the vegetation period in the southern part of the country has been increasing, the length of the vegetation period in northern Sweden is estimated to increase about 14 days in the past 40 years with a rapid rise recorded in the last decade (smhi.se 2019).

The length of the vegetation period is expected to be very long by the similar number of days in spring and in autumn with the more prominent variation to be observed in the South, and weaker changes to be recorded in the North of Sweden.

#### 2.4.4 Windstorms

The most anticipated extreme weather events are storms which unpredictably affect Swedish forests. See, for example, the winter storm Gudrun that hit the country in 2005 (Haanpää *et al.2005*). At the same magnitude or more severe windstorms, the risk of wind-felling of the forest will always be higher due to weather conditions and seasonal soil properties.

The primary driver of forest damages is wind speed. The critical limit of average wind speed to cause significant adverse effects on the forest is set to be about 15 m/s and blasts of about 30 m/s (Kellomäki et al., 2005). In Sweden, surprisingly the blasts of Gudrun storm reached more than 40 m/s on the west coast and almost the same speed in the central parts of the country.

More importantly, different studies found that the dominating planted Norway spruce was much more vulnerable to storm winds, particularly with massive falling trees of between 30-40 years of age (e.g., Alexandersson *et al.* 2005)

Also, several other factors affect the robustness of forests. The prime factors are the thinning and harvesting methods since a recently and heavily thinned forest stand will be damaged easily by a wind storm (Kellomäki et al., 2005). In the last ten decades, the area of forest plantations has increased substantially, and the earliest tree stands as well as grasslands have become less steady in soils. On top of that, the current massive heavy harvesting equipment put undoubtedly the value of toppling timber at high risk (Alexandersson, *et al.* 2005). Therefore, indeed, human activities have also increased the risk of severe impacts significantly due to storm events towards forest stands.

## 3 Literature review

Effects of intensive silvicultural practices and weather patterns on Swedish forests are expected to be tremendous. The complex variability of weather conditions may generate positive as well as negative effects on forest growth over an extended period and in different geographical locations. Therefore, in the broad interests of forest industry profitability, applying intensive forest management systems needs to minimize the risk of adverse effects and maximize the benefits of weather variations.

In Sweden, forest growth has been at an increasingly higher rate compared to the annual forest harvesting volume over the last 100 years. Furthermore, the effects of intensive silvicultural methods to increase forest standing volume have contributed importantly to a more massive potential increase of timber harvest as well as to increased carbon sequestration provided by the forest ecosystems (SFA 2015). For example, the results from the empirical study suggest that forest management systems with the purpose to increase biomass productivity in the boreal zones would contribute to a reduction of further carbon emissions by about 40 million tons of CO2 emissions annually (Lundmark *et al.* 2014). The present study further tests this hypothesis by introducing silvicultural treatments and weather variability as determinants of growth rates in Swedish forests.

### 3.1 Management practices and forest growth

The optimal forest management practices that determine the level of carbon sequestration are relatively dubious in the long run. Many studies suggest that the optimal rotation period for clear-cutting forest stands and planting new seedlings lead to a more substantial gain of carbon storage in the context of wood bioenergy production because an increase of forest biomass is more likely in young forests (Sedjo & Tian 2012). However, there are other secondary drivers such as global warming that may affect forests to react as net emitters rather than carbon emissions sinkers which could shorten the optimal rotation period of forest stands (Bellassen & Luyssaert 2014).

Low availability of mineral fertilization in boreal regions is a limiting factor on the growth rate (Högberg *et al.* 2017). Several studies confirmed that fertilization added nutrients that have a tremendous impact on tree growth (Fox *et al.* 2007; Saarsalmi *et al.* 2014; Noormets *et al.* 2015). Additionally, it is recommended to fertilize the later forest stands rotation because this fertilization timing leads to the productivity of high-quality timber rather than improving the branches of the butt log before clear felling (Tapio 2013). Also, when mineral soils of the boreal zone suffer a lack of nutrients, the critical recommendation is fertilizing (Saarsalmi & Tamminen 2005).

Many series of experiments focused chiefly on the fertilization effects for the present rotation period, but it turns out that these effects exist until the next period. For instance, a large quantity of pre-harvest fertilizers has adverse effects on soil chemistry after final harvest (Ring 2004). Some contradicting empirical results which found positive effects are, however, also reported (Johansson *et al.* 2013b). Similarly, another experimental study that estimated the long-term effects of fertilization regimes found the growth of stem-wood increments to be significant with intensive fertilization. The growth response relationship was, however, relatively robust in the less intensity fertilization regimes (Jacobson & Pettersson 2010).

Based on modern technologies, experimental studies were carried out to understand the apparent effects of thinning on both dominating tree species in Sweden, Norway spruces, and Scots pine. Some of the tested hypotheses were to examine whether management treatments that associated thinning with use of fertilizers could have beneficial effects on forest growth and yield. More profoundly, these studies intended to understand the underlying effects on tree volume growth, the stability of forest stands, and the quality of woods under separate thinning periods at different intensity (Eriksson 2006). The results are exciting but also conflicting. He found that management practices that intended to maximize biomass production could, in turn, have adverse effects on maximizing carbon sequestration. The study results asserted that thinning regimes for spruce stands have no significant effects on mean annual increment, and fertilizers alone also show no positive effects on biomass production when applied from above-ground. Accordingly, combining thinning and fertilization in the pine blocks showed significant results with an increase in the mean annual increment for biofuel production.

He concluded that when the objective is to maximize the biomass productivity, the enormous standing volume for both species is observed under the unthinned regime and thereby increasing carbon storage.

Scarification practices improve the planted seedlings' growth as an effect of the changes in the environment, such as the availability of nutrients and soil compositions, i.e., temperature (Johansson *et al.* 2006). The great prospect of scarification is to eliminate competing vegetation, which in turn more facilitates newly planted seedlings to grow under favourable conditions than woody and other herbaceous species of which can typically dominate the area after harvesting (Thiffault *et al.* 2005). Several other studies have found that scarification contributes positively over time. For example, scarification enhances forest stand to be homogenous and preserves the initial gains for planted seedlings for a very long time (Nilsson & Allen 2003; Johansson *et al.* 2013b). Accordingly, scarification is not only carried out primarily to promote seedlings growth but also to protect these seedlings against diseases such as pine weevils (Hylovius abietis) (Nordlander *et al.* 2011). Therefore, the importance of scarification in reducing the pine weevil attacks will likely be significant in the future because the current forest protection forbids the use of insecticides.

In recent years, scarification has gained much interest due to technological progress in the forest sector in Sweden (Johansson *et al.* 2013a). A limited number of studies have asserted a positive relationship between forest productivity and intensive scarification for better management of forests (Mattsson & Bergsten, 2003; Nordborg *et al.* 2006). As a result, the corresponding scarification cost could make the entire forest project more expensive during pre-commercial thinning even though it raises tree biomass increment. Conversely, some of its long-term effects could lead to much more resource nutrients, which in return contribute undoubtedly to the non-declining height as well as the basal area of young planted trees. For instance, (Johansson *et al.* 2013a) found that there was a remarkable difference in tree heights over the first 14 years between forest stands with scarification treatments and non-treatment stands. After then, trees follow a constant growth rate from 14 to 18 years of age of which the difference from both standing trees maintained uninterruptedly, and growth curves stopped to diverge. It turned out that forest sites located farthest North indicated a vast difference in growth heights which could reach up to 11 years, and this incredible time difference represents the enormous effect on the economy of the forestry industry.

### 3.2 Weather patterns and forest growth

A study that assessed the effects of climate variability on the function and structure of boreal coniferous forests found that an increase in temperature could enhance forest function substantially (Kellomäki & Väisänen 1997). The results show tremendous effects on forest growth when the temperature is associated with an elevation of carbon from above-ground. They reported that forest growth increases at lower levels than expected because of the higher resistance of trees during elevated temperature. However, this result follows the underlying assumption that respiration during dark periods is lowered to no considerable extent by elevation of CO2 concentration.

The boreal coniferous forests are susceptible to high temperature. Using the process-based simulation model to estimate and compare the effect of elevated temperature and CO2 concentration on net primary production (NPP) in the Nordic countries (Bergh *et al.* 2003)), they found an increase of about 24-37% of NPP in spring for Scots pine and Norway spruce. However, beech stands did not respond conveniently with the increase in temperature of spring to the same extent as the other deciduous species. In Denmark, for example, the effect of elevated CO2 concentration was more observed for the beech stands. The high CO2 concentration helps the efficient use of water in stands with water deficit.

Different reasons can explain a positive relationship between average temperatures and the growth in October of the previous year. Higher October temperatures are likely to contribute to the extension period of physiological function until the end of the growing season. Thus, allowing exhausting relocation of transportable nutrients from leaves to seasonal living parts of the trees during the prolonged time. For instance, (Barbaroux & Breda 2002) reported that timber maximizes the reserves storage in October. Besides, a similar study of red oak (Q. rubra L.) revealed the maximum sugars relocated to the main stem existed in late October (Xu & Griffin 2006). Therefore, October reveals warmer weather conditions which determine the amount of the carbohydrates transferred to perennial parts of the tree and thus producing more available nutrients and energy reserves for the next growing season.

On the other hand, the damage of severe windstorms had significantly increased both in frequency and magnitude during the later 20<sup>th</sup> century (Munich Re 2002).

There are several other instruments such as changes in stand age and structure, and forestry management treatments that also vary over time, consequently reshaping windstorms sensitivity to forests. An analysis of the panel data on windstorm damage in Sweden during 1901-2000 Nilsson *et al.* (2004) found that storm damage to forest frequently manifested at the age of 51 years. An empirical assumption is that thinning forest stands will need to be done at the age of 41 years and that the stand height is significant enough to imply being exposed to the risk of windthrow (Persson, 1975). During that period, the study estimated 77 different windstorms and a total volume loss of about 110.7 million m<sup>3</sup> (Nilsson *et al.* 2004). It appears that the terrible storm damage was observed more in the south than in the north of Sweden. The results also indicate that the increase in storm damages in the entire country over the last century was likely due to the relatively larger forest proportion in the landscape and absolute volume associated with the age classes inclined to storm damage, not because of storm frequency and its magnitude (Nilsson *et al.* 2004). Also, the changes in management practices may be another substantial cause.

#### 3.3 Carbon Sequestration

The age of the forest is one of the first measurements that help to quantify the amount of carbon sequestration. The growth of tree biomass rises rapidly in the early stage of plantation following a non-linear growth rate, unlike the flowing of carbon that has mainly been estimated using linear functions without considering forest age. As a result, the studies that included the age of forest while estimating non-linear tree growth function noted minor carbon emissions sink (Brainard et al. 2009). More economic factors and biological properties of forests are needed to predict accurate results. Like in most other countries, there is no recent market valuation of carbon sequestration in Sweden (Gren & Aklilu 2016; Gren & Amuakwa-Mensah 2018). Estimating the unit of the carbon sequestration needs to compare the related costs of reaching climate neutrality through the reduction of carbon emissions and any other greenhouse gases (GHG). In this study, the current tax scheme on carbon emissions is used to determine the value of carbon sequestration as the societal willingness to reduce GHG emissions. For example, in a thorough study conducted in the Swedish forests (SEPA 2012), it was found that the annual carbon emissions sink could reach up to 38 million tonnes of CO2 emissions per year. Forests with the ability to grow further are the net saver of CO2 emissions and forest management practices have vital effects on carbon sequestration at a larger scale.

## 4 Methodology and empirical data

This section illustrates the methodology of fully modified ordinary least square (FMOLS), and the empirical work of creating an estimation of the total economic value (TEV) of the forest ecosystem services of timber and carbon sequestration. Also, this includes describing unit net prices that were used in this model.

### 4.1 Methodology

Two main steps are conducted to produce the total economic value of forest ecosystem services focusing mostly on the value of timber and carbon sequestration. The starting point is a step (i) which examines effects of weather patterns and management systems on forest growth, and next step (ii) is the estimation of the total economic value of timber production and carbon sequestration.

In forestry, non-linear growth functions are mostly preferred compared to linear modeling since the linear functions often fail to deliver according to the educated guesses (Wang 2006). However, Hastie & Tibshirani (1986) and Wood (2017) suggested that using additive models' approach is essential because it handles complex relationships between the explanatory variables and the response. The recommendation from economic literature is that step (i), of estimating the effects of weather variability and forest management practices on forest productivity, is regarded as estimating the growth function. It utilizes an underlying assumption that the growth function follows a sigmoid shape from the starting point (0,0) which is then followed by inflection point that appears at the beginning of the adolescent stage, and reaches the peaks and then falls in the maturity stage (Fekedulegn *et al.* 1999).

The non-linear relationships with their rigorous mathematical attributes pass the fitness test by providing meaningful feedback about the rate of growth, the inflection point, the initial forest stands, and the maximum yield. Nonetheless, there is a high correlation of the parameters in non-linear functions because a change in one estimated parameter induces changes in other parameters to keep its original functional form. On the other hand, using the adding-up approach, Amuakwa-Mensah *et al.* (2016) concluded that explanatory variables could lead to a wrong conclusion about the relationship between these variables and the response.

Therefore, this study selects the logistic function in the form of a linear transformation because it is more pertinent for estimating the parameters of the model than using other non-linear growth functions such as Von Bertalanffy, and Schnute (Fekedulegn *et al.* 1999). Besides that, the linearized growth function has illustrated to providing more accurate predictions (Nguimkeu 2014). Accordingly, this growth model usually tends to be a beneficial instrument for studying the tendencies of different population growth Nguimkeu & Rekkas (2011) such as the forest in applied research.

Given that this thesis opts to use a linearized logistic function with no intercept, it allows equation (1) to be expressed as follows;

$$y_{t+1} = y_t \left( 1 + \beta_1 \left( 1 - \frac{y_t}{\beta_2} \right) \right) - H_t + \beta_3 X_t + \beta_4 W_t$$
(1)

It is expressing the increase in the volume of forest growth per hectare during year  $(y_{t+1})$  as a function of forest volume in the previous year  $(y_t)$ , the harvest of timber  $(H_t)$ , a vector of intensive management practices  $(X_t)$  and weather variability  $(W_t)$ . A linear logistic function expressed in terms of intrinsic growth rate,  $\beta_1$ , and maximum productivity,  $\beta_2$  helps to estimate impacts of a marginal change in management and weather on the forest growth. Moving the annual harvest  $(H_t)$  to the left-hand side will minimize the problem of multicollinearity between volume harvest and a set of management treatments. By dividing both sides by the standing volume per unit area  $(y_{it})$  and keeping the  $(W_t)$  aside off the equation (1) is transformed into equation (2).

$$\dot{Y} = \beta_1 \left( 1 - \frac{y_t}{\beta_2} \right) + \beta_3 \frac{x_t}{y_t} \tag{2}$$

Where 
$$\dot{Y} = \frac{y_{t+1} - y_t + H_t}{y_t}$$
 represents 'adjusted growth rate'. (3)

The adjusted growth rate is the value of the growth of forest productivity. With direct effects of intensive management practices involving thinning, scarification and fertilization and weather variability which includes variations in temperature, precipitation, length of the vegetation periods and windstorms, I expect the intrinsic growth rate, not least  $\beta_1$  to be positive but it is arduous predicting the impact of weather variability, not least the sign of  $\beta_4$ .

The marginal impact of intensive silvicultural systems on the forest growth rate may also be ambiguous, not least the sign of  $\beta$ 3. While estimating the impact of weather patterns and silvicultural practices, the response variable is narrowed to the rate of growth in forest productivity over the years, equation (3).

The statistical results on the data used in the econometric analysis are presented in table 1.

|   |     |         |          |         | -        |
|---|-----|---------|----------|---------|----------|
| Variables                                       | Ν   | Mean    | Std. Dev | min     | max      |
| $\dot{Y}_t$ : Adjusted growth rate <sup>a</sup> | 195 | 0.0368  | 0.0670   | -0.3721 | 0.5069   |
| $H_t$ : Harvest, $m^3/ha$                       | 196 | 2.5645  | 1.3694   | 0.4849  | 7.8676   |
| $y_t$ : Standing volume, $m^3/ha$               | 196 | 93.8149 | 32.0559  | 29.3409 | 163.7074 |
| Scarification, % of productive area             | 196 | 0.00625 | 0.0019   | 0.0174  | 0.0114   |
| Thinning, % of productive area                  | 196 | 0.01278 | 0.0045   | 0.0056  | 0.0277   |
| Fertilization, % of productive area             | 196 | 0.0035  | 0.0031   | 0       | 0.0132   |
| Fertilization policy                            | 193 | 0.1011  | 0.1703   | 0       | 0.8847   |
| Temperature, degree Celsius                     | 196 | 0.4210  | 1.0217   | -2.05   | 2.3      |
| Precipitation, mm                               | 196 | 4.9948  | 14.911   | -29     | 47.6     |
| Windstorm, m/s                                  | 196 | 2.376   | 1.1577   | 0.48    | 6.32     |
| Length of the vegetation period,                | 196 | 4.2925  | 7.6048   | -11.78  | 26.62    |
| number of days.                                 |     |         |          |         |          |

#### **Table 1. Descriptive statistics**

<sup>a</sup> 
$$\dot{Y}_t = \frac{y_{t+1} - y_t + H_t}{y_t}$$
, the adjusted growth rate.

Also, introducing the indirect effects of weather patterns and management practices as explanatory variables on the intrinsic growth rate following the (Amuakwa-Mensah *et al.2016*; Gren *et al.* 2016) gives the equation (4)

$$\dot{Y}_{t} = \left(\theta_{i1} + \theta_{i2}X_{it} + \theta_{i3}W_{it}\right) \left(1 - \frac{y_{it}}{\beta_{i2}}\right)$$
(4)

where  $\beta_{it1} = \theta_{i1} + \theta_{i2}X_{it} + \theta_{i3}W_{it}$ . This regression function corresponds to equation (5)

$$\dot{Y}_{t} = \rho_{i1} + \rho_{i2} y_{it} + \rho_{i3} X_{it} + \rho_{i4} W_{it} + \varepsilon_{it}$$
(5)

 $\rho_{i1} = \theta_{i1}$ , and  $\rho_{i2} = -\frac{\beta_{i1}}{\beta_{i2}}$ . Now the intrinsic growth rate has changed to time dependent and  $\beta_{it2} = \rho_{i1} + \rho_{i3}X_{it} + \rho_{i4}W_{it}$ .

The rest of the parameters are like the ones in the direct model. However, this study does not analysis the results of the indirect approach.

Finally, in step (ii) changes in forest productivity affect the total economic value through timber harvesting and carbon sequestration.

Quantification of the monetary value using the unit net income from timber and carbon sequestration,  $P^T$  and  $P^{CO2}$  respectively, implies that the total monetary benefits that society gains from the two provision types of forest ecosystems is determined in the equation (6).

$$TEV = P^{T} * H_{t} + P^{CO2}Conv(y_{(t+1)} - y_{t})$$
(6)

 $P^{T}$  is the price per cubic meter, H<sub>t</sub> is the volume harvested in cubic meter (m<sup>3</sup>) and *conv* is the conversion of change in forest volume to carbon dioxide equivalents (CO<sub>2</sub>e) and needs to be calculated.

The fully modified ordinary least squares (FMOLS) approach (Wang & Wu 2012) is used in this study to estimate the type of equation (6). It follows that the fully modified OLS as proposed by Phillips & Hansen (1990) tends to be a half parametric model and is robust to endogeneity, though with the presence of serial correlation problems. Furthermore, FMOLS produces coherent and adequate estimations for the model lacking cointegration relationship and implies outstanding robustness for both stationary and non-stationary series in the same cointegration analysis. Since a long panel data is used for the analysis of this thesis, FMOLS becomes very handy to avoid problems related to serial correlation and non-stationarity. Using Stata, the system starts by modifying variables and is followed directly by discarding the existing undesirable parameters. The FMOLS is structured in a way that allows for correcting endogeneity and serial correlation problems using an adequate approach in the sense that it maintains the original functional form of the linearized logistic function.

### 4.2 Empirical data .

This thesis uses panel data to estimate the forest productivity as described in chapter 4.1, on four regions of Sweden for the period 1965-2013. The data on forest related indicators is provided from the Swedish Forestry Agency 2014 and weather conditions are obtained from the Swedish Meteorological and Hydrological Institute (SMHI) 2017. "The materials cover all statistics during the entire study period, and they are deviations from the annual average values for forest management practices and weighted yearly average values for weather variability."

Measurements of scarification, thinning and fertilization are in terms of percentage of forest productive areas and the indicator of harvest is in cubic meter per hectare. The total average standing volume in all regions of Sweden amounts approximately 120 m<sup>3</sup>/ha, and the southern parts comprise the highest standing volume, whereas the northern parts aggregate the lowest volume. The total standing volume was 3 billion cubic meters of productive forest land area (Christiansen & Fong Ekistrand 2013). The four different forest zones have different climate conditions (fig. A1). The northern regions comprise of Norra and Södra Norrlands and are the largest zones with more significant forest area compared to the southern regions, not least Svealand and Götaland. The larger part of Swedish forest is the boreal coniferous forest with only two dominating tree species, Scots pine (Pinus sylvestris) and Norwegian spruce (Picea abies) in the entire country totalizing between 76% and 84% of standing volume in productive forest area (Christiansen & Fong Ekistrand 2014). The rest of tree species which mostly are broad-leaved trees are found mainly in the southern region. Birch (Betula spp.) takes about 10-15% of the total standing volume, and beech (Fagus spp.) and oak (Quercus spp.) both combined comprise nearly 6% of the total standing volume and most are in Götaland. The standing volume measured in m<sup>3</sup>/ha differs in all four forest regions (Amuakwa-Mensah et al.2016). The standing volume in Götaland almost doubles the standing volume in Norra Norrland. Several reports have found that there has been a steady increase in standing volume per hectare in a productive forest area during the same period (Amuakwa-Mensah et al. 2016; Christiansen & Fong Ekistrand 2014).

In 1985 a policy of regulating fertilizers was introduced in the forestry sector due to the eutrophication problem of coastal waters in Sweden. This problem is linked to the overload of nutrients which lead to damaging the bottom sea areas then creating an increased frequency of toxic blue algae while also changing fisheries composition at the cost of commercial species (Conley *et al.* 2009). These sorts of damages hit Götaland severely, and therefore, authorities imposed stricter regulations in this region than in the other parts of Sweden. Since then, data collection was also changed in the same year as the introduction of fertilization policy. This empirical study introduces a dummy value which takes a value of 1 from 1985 onwards and 0 otherwise with the purpose to maintain impacts of the fertilization policy and change in data collection on the growth rate.

On the other hand, this study uses the weighted average values of the maximum and minimum temperatures while considering the temperature at given moments in all four different regions.

The observations provide an almost correct, and accurate yearly weighted average temperature that follows the international rules. They are calculated based on the average temperature of the twelve rounded monthly average values rather than the daily average values, which are recorded every three hours a day on the Celsius scale (smhi.se 2018). Also, the annual weighted average temperature is rounded to one decimal of degree Celsius. Similarly, windstorm, which is known to be the natural event, is measured in meters per second using the annual weighted mean value of the wind following the principle of the average wind values recorded for every ten minutes in Sweden (smhi.se).

Precipitation in Sweden is one of the vital meteorological parameters, which is measured in millimetres (mm). This research uses the yearly weighted average values of the rainfall. The precipitation provides a 1 mm thick layer on a horizontal surface with no presence of actual evaporation and water flowing away. However, deviations from the mean values of precipitation do not take into account the amount of moisture and hoarfrost as in other parts of the world. Differently, the length of the vegetation period with a daily mean temperature exceeding a certain threshold indicates the time of the year for the tree to grow and is usually measured in terms of a period of time such as days or weeks (smhi.se 2019). Since the starting of this growing season chiefly relies on the air changing temperatures, and other combing factors ambiguously explain the end of the season such as prolonged daylight time, conditions of moisture, utmost temperatures, extreme frost occurrence, etc may also explain the minimum value of the length of the vegetation period to have a negative sign as indicated in the table of description (table 1).

The underlying assumption that forest owners are rational and they should optimize the value of timber to be harvested in each period, H<sub>t</sub>, based on the market prices and forest growth (e.g., (Sohngen & Mendelsohn 2003). A further assumption is that carbon sequestration is part and parcel of harvesting decision. Therefore, forest carbon sequestration uses the existing payment model to reduce the emissions of greenhouse gases (GHG). There are several studies on estimating the forest carbon sequestration and the impacts of management systems (Janssens *et al.* 2005; Helin *et al.* 2013; Lundmark *et al.* 2014). This study evaluates carbon located in the aboveground biomass and ignores carbon found in soil. However, the total felling and wood made materials cause the release of carbon sequestration in the Swedish forests concluded that the averaged value had reached 0.47 ton  $CO_2e/m^3$  harvest (Lundmark *et al.* 2014).

The present study, however, does not consider the same approach since there is a lack of data on carbon sequestration from the growth rate. Instead, it is calculated. The use of the conversion for felling, which means less carbon sequestration per hectare because of the average growth rate in standing volume is higher compared to the average harvest.

## 5 Results analysis

The four forest zones provided information that was useful in forming a panel data to be used for estimating the impact of management and weather on forest productivity for the entire country. In this study, the panel data gives adequate observations with no further adjustments to get the accuracy and efficiency of the regression estimates. Furthermore, introducing zonal fixed effect helps to characterize each zone with unique features because these specific features may affect the response variable.

Thus, having presented data in the previous chapters and based on the econometric models, the results are analyzed following the estimated forest productivity in association with total economic value computations. The use of logistic growth functions with parameter values which estimate the effect of management practices and weather patterns turn out to yield significant productivity. It was decided to test the effect of lagged and non-lagged variables of these practices and weather variability because of delays in the effect of both explanatory variables. As such, the results indicated that the lagged variables produce a better fit of statistical significance compared to the findings from non-lagged model. Thus, all regression estimates in this study consider the lagged variables. See the results of the logistic functions, as presented in table 2.

| VARIABLES                              | Direct     | Indirect   |
|--|------------|------------|
|  | Model      | Model      |
|  |            |            |
| Constant                               | 0.1012***  | 0.1307***  |
|  | (0.0168)   | (0.0133)   |
| Standing per hectare                   | -2.9421*** | -2.7124*** |
|  | (0.3349)   | (0.3276)   |
| Fertilizer (_1) <sup>g</sup>           | 0.0660     | -1.9036    |
|  | (0.0926)   | (1.4252)   |
| Fertilization policy (_1) <sup>g</sup> | 0.0001     | 0.0322     |
|  | (0.0016)   | (0.0231)   |
| Thinning (_1) <sup>g</sup>             | -0.0694    | -0.0482    |
|  | (0.0501)   | (0.6605)   |
| Scarification (_1) <sup>g</sup>        | 0.3906***  | 2.3375     |
|  | (0.1186)   | (1.4911)   |
| Temperature (_1) <sup>g</sup>          | 0.0079***  | 0.0079***  |
|  | (0.0020)   | (0.0021)   |
| Södra Norrland                         | 0.0383***  | 0.0357**   |
|  | (0.0146)   | (0.0140)   |
| Svealand                               | 0.0077     | 0.0177     |
|  | (0.0286)   | (0.0281)   |
| Götaland                               | 0.0234     | 0.0411     |
|  | (0.0403)   | (0.0399)   |
| Trend                                  | 0.0017***  | 0.0012***  |
|  | (0.0004)   | (0.0004)   |
| Precipitation (_1) <sup>g</sup>        | 0.0001     | 0.0001     |
|  | (0.0001)   | (0.0001)   |
| Length vegetation (_1) <sup>g</sup>    | 0.0011***  | 0.0009***  |
|  | (0.0003)   | (0.0003)   |
| Windstorm (_1) <sup>g</sup>            | 0.0010     | 0.0009     |
|  | (0.0016)   | (0.0016)   |
| Observations                           | 187        | 187        |
| R <sup>2</sup>                         | 0.0526     | 0.0718     |
| Long Run SE                            | 0.0248     | 0.0247     |
| Bandwidth(neweywest)                   | 65.02      | 67.40      |
|  |            |            |

Table 2: Regression results from fully modified ordinary least square (FMOLS) of the growth rate of Swedish forest productivity (Standard errors in brackets).

<sup>g</sup> One-period lag. Significance level: \*\*\* p<0.01, \*\* p<0.05, \* p<0.1 The direct and the indirect regression models with the management of scarification and weather patterns of temperature and the length of the vegetation period indicate to have a better fit of statistical significance. Both models retain small and almost equal long-run standard errors, which is 0.0248 for the direct model and 0.0247 for the indirect model. The analysis of this study chooses to use the direct model because it provides straight forward results and easy to interpret.

These growth models generate the constant of intrinsic growth rate  $\beta_1$ , which is significant in both the direct and the indirect models, with the estimates of 0.10 and 0.13, respectively. The estimated constant growth rate of this study is closer to the amount of 0.13 found in the direct model of the assessment of the impacts of site quality on growth rate and variability of forest productivity in Sweden Amuakwa-Mensah *et al.* (2018). Additionally, this intrinsic growth rate is convenient to the range of the estimate of 0.11 reported in the boreal forest at large scale (Eriksson & Vesterberg 2016). Also, the standing volume retains the negative sign as expected, and it is reported to be -2.94, and it is closer to the estimate of -2.10 found in the study of the effects of the site quality on the growth rate in Sweden *Amuakwa-Mensah et al.* (2018). In other words, the growth rate in forest productivity is reported to be at a low level with high productivity degree. The indicator signaling variations in standing volume per hectare productive area over time, the variable of the trend, for the period 1965-2013 is recorded to be positive and significant. The regression equation reported the trend to be 0.0017 and 0.0012 in the direct and indirect model, respectively, which means that all four forest regions experienced a steady increase in standing volume.

### 5.1 Management practices and forest productivity.

The regression results pass the fitness test while assessing the effects of management practices on forest productivity. It turned out that scarification is the only management treatment which has a significant effect on productivity, whereas thinning and fertilization practices have no significant effects on forest growth. Scarification is the most pertaining practice to enhance environmental conditions which favor forest plantations in Sweden. With the direct model, the estimated impact of scarification is positive and equals 0.39. Which means that a one-unit increase in scarification contributes about 0.39 percentage point in forest productivity. One important reason could be that this management treatment not only contributes significantly to the increment of the standing volume per ha but also at high levels of productivity. Thus, the growth rate increases with the best scarification practice and at the same time, which leads to lower mortality of planted trees. Scarification, therefore, would be expected to have a significant effect on the economic value of the forest stand. Another reason could be that the one-period lagged model of these management indicators succeeds in capturing only the long-term effect of scarification.

When considering the indirect impacts, there is no single management system which shows significant effect on the growth rate. The practices of fertilization and thinning are negative, and they reveal to have no significant effects on forest productivity. Conversely, scarification and fertilization policy are positive, but they also have no important effects on productivity. The reason might be that since these practices are positive, they may have significant effects on the growth of the standing volume per ha with a low degree at higher productivity. Which means that the growth rate increases as the magnitude of the management practices also increase. Furthermore, it could be that the one-period lag of these variables does not consider the long-term impacts of the management methods in the model.

## 5.2 Weather variability and forest productivity

Weather variability in boreal forests exhibits attributes on standing volume due to relatively annual growth, warmer temperatures, and the number of daily degrees. The findings of this study demonstrate that the changes in some weather patterns can have profound effects on forest ecosystem services in Sweden. Temperature and length of the vegetation period are both positive and significant at 1% level. Also, they reveal to have a considerable impact through the increase of the expected economic benefit of carbon sequestration with the estimates of 0.0079 and 0.0011, respectively. In the long run, one of the critical reasons could be that on average, high temperature induces the increase of the stem-wood growth of planted trees in cold regions (i.e., boreal zones). The temperature is vital for metabolic processes entailing storage of carbon, which is an essential factor to aboveground plant productivity. Therefore, the rise of temperature together with adequate precipitation and nutrients help in the accumulation of higher biomass.

Furthermore, one of the indirect impacts of warming is often the length of the vegetation period, also known as the growing season. The results show that the length of the growing season has increased over the past decades, and it has a positive effect on the forest growth rate and at the same time on the increase of carbon sequestrated. Unfortunately, these results cannot be compared to any other study since there has been hitherto no similar research.

Similarly, the factors of temperature and the length of the vegetation period in the indirect model have closer significant impacts on the forest growth, with the estimates of 0.0079 and 0.0009, respectively. On the other hand, patterns of precipitation and windstorm reveal to have no significant impacts on forest growth. Decreasing of summer precipitation may cause soil water shortages in Swedish forests, and these shortages could be due to the increase in temperatures. The indicator for precipitation shows no effect on forest productivity since it is estimated to be zero, 0.0001, while the estimate for windstorm is 0,0010, and both variables are not significant.

### 5.3 Estimation of the total economic value

All monetary values are expressed in SEK. The price of timber is SEK  $455/m^3$  for saw timber and SEK  $284/m^3$  for pulpwood. While the harvested volume used for saw timber and pulpwood reached nearly about 40% and 50% respectively (Nilsson *et al.* 2017), the remaining wood materials are most important for bioenergy, and there is no official data on prices because forest owners consume them locally (Carlsson 2012). Thus, this study uses the average price for timber,  $P^T$ , of SEK 360/m<sup>3</sup> (Gren & Amuakwa-Mensah 2018).

With regards to the unit price of carbon sequestration i.e.,  $P^{CO2}$ , there is extensive body of literature that analyzes the marginal cost of carbon for society. They have reported to range from 50 and 2470 SEK/ton CO2e (a review (Tol 2013)). In the recent study that takes into account the risk of uncertainty, the value of carbon sink enhancement could reach up to 2540/ha (Gren & Amuakwa-Mensah, 2018). The cost of carbon on the European trading market could change continuously and range between 45 and 300 SEK/ton CO2e from 2007-2014 (MacDonald 2016).This thesis has considered the average carbon price of SEK 162 /ton CO2ev/m<sup>3</sup>.

Furthermore, the conversion factor of forest biomass, *Conv* is 0.91 ton CO2e (Carbon dioxide equivalents) /m<sup>3</sup> harvest volume for coniferous and 1.47 ton CO2e/m<sup>3</sup> for deciduous forest (Vass & Elofsson 2016; Gren & Amuakwa-Mensah 2018). Since the conifer and deciduous forest cover areas are equivalent to 89% and 11% respectively. It appears that the conversion factor for average carbon sequestration is then conv=0.91\*89%+1.47\*11%=0.97-ton CO2e/m<sup>3</sup>.

Having parameterized the growth rate, we can now estimate the values of timber and carbon sequestration based on the average values of parameters of interest. The total economic benefit of these ecosystems is calculated following equation (6) and the principles and data, as explained in previous sections of chapter 4. The calculation of the contribution of significant weather variability to the increase of monetary benefit gives the total economic value of forest ecosystems. At start, the total economic value is calculated for the base scenario, whereby no other consideration of variable changes. After that, temperature and the length of the vegetation period are introduced in the calculations since they appear to be significant for forest ecosystems studied.

Having  $\frac{\Delta \dot{Y}_t}{\Delta t emperature} = 0.0079$ ; the change in adjusted growth rate relative to the change in temperatures is constant and equals the coefficient value of temperatures. Therefore, the new  $\dot{Y}_t = average \ \dot{Y}_t + \frac{\Delta \dot{Y}_t}{\Delta t emperature}$ ; a one-unit increment of temperature will generate a new adjusted growth rate starting from the levels of average growth rate.

From equation (3), the newly adjusted growth rate amounts to

 $\frac{y_{(t+1)-}y_t + H_t}{y_t} = \dot{Y}_t + \frac{\Delta \dot{Y}_t}{\Delta t emperature}; \text{ which gives the difference in variation of standing volumes}$ in different periods. Thus,  $y_{(t+1)} - y_t = y_t (\dot{Y}_t + 0.0079) - H_t.$ 

In the similar procedures, a one-unit change of length of the vegetation period will impact the growth rate as follows;  $y_{(t+1)} - y_t = y_t (\dot{Y}_t + 0.0011) - H_t$ . Replacing the equalities of differences of standing volumes per hectare in the equation (6) produces the table (3) and it shows the total economic benefits of timber and carbon sequestration in various scenarios.

This means that,  $TEV_{Temp.} = P^T * H_t + P^{CO2}Conv[y_t(\dot{Y}_t + 0.0079) - H_t]$  is due to changes in temperature and  $TEV_{LengthVeg.} = P^T * H_t + P^{CO2}Conv[y_t(\dot{Y}_t + 0.0011) - H_t]$  follows after the increase of the length of the vegetation period.

Table 3: Calculated total economic value and the average contribution of a one-unit increase of temperature and length of the vegetation period to the total economic value of timber and carbon sequestration in SEK/m3/ha.

|                       | A one-unit increment of: |             |                          |  |
|-----------------------|--------------------------|-------------|--------------------------|--|
|                       | Base Scenario            | Temperature | Length vegetation period |  |
| Timber:               | 923.22                   | 923.22      | 923.22                   |  |
| Carbon Sequestration: | 141.55                   | 258.01      | 157.76                   |  |
| Total:                | 1 064.77                 | 1 181.23    | 1 080.99                 |  |

The results show that the expected economic benefit of the carbon sequestration augmented at different levels and the value of timber remained constant. These changes in carbon sink constitute the total impact of weather patterns on forest ecosystem services in Sweden. Furthermore, these variations are the same for all the regions under this study since the trend shows to have a positive and significant effect.

## 6 Discussion

The results indicate that the value of timber remains unchanged in all calculations and amounts to SEK 923 m<sup>3</sup>/ha since the static model used in this study does not allow the significant temperature and the length of the vegetation period having effects on timber production. On the other hand, the model permits the same significant variables to have a considerable impact on the value of carbon sequestration. The expected benefits of carbon sequestration with a one-unit increase in temperature augmented to SEK 258/ton CO2e/m3 from the initial benefit of SEK 142/ton CO2e/m3. Moreover, with a one-unit increase of the length of the vegetation period, the value of carbon increases by 11.5%, which means that the value reaches SEK 158/ton CO2e/m<sup>3</sup>. The total economic benefit in the base scenario is estimated to be SEK 1065 /m<sup>3</sup>/ha. The rise of temperature and the length of the vegetation period increases the total economic value by 11 and 1.5 percent to attain SEK 1181 /m<sup>3</sup>/ha and SEK 1081 /m<sup>3</sup>/ha respectively compared to the base scenario. The increase is mainly due to the changes in carbon sequestration.

The calculation of the total expected monetary benefit of carbon sequestration in the whole country would be based on the estimation of total standing volume of productive forest land, which is reported to be 3 billion cubic meters (Christiansen & Fong Ekistrand 2014). However, since weather conditions are different across the regions, it cannot be easily possible to scale up the value of carbon sequestration at the national level (Amuakwa-Mensah 2019). On the other hand, following the unit price of timber as estimated in the table (3) the total economic benefit of timber production for the entire country equals 108.00 10<sup>10</sup> SEK (see table A2). The results show that the economic benefit of this forest ecosystem is still constant and corresponds to 23% of the gross domestic product (GDP) in 2018. Reducing the value of timber production to 300 SEK/m3 its economic benefit drops to 19 % of the gross domestic product (GDP) in the same year. The comparison with the GDP is based on the data provided by (World Bank 2019) and the exchange rate of December 2018 (*Riksbank. Statistics* 2019). (See table A3)

The calculated economic values rely heavily on the prevailing assumptions of the price of timber and carbon sequestration. The values of the parameters used in the evaluation of table (3) rest unchanged. The average price of timber is  $360 \text{ SEK/m}^3$  during the period of the study.

The reference estimate of the carbon emissions tax is the average of the carbon price and amounted 162 SEK/ton CO2e/m<sup>3</sup>. Also, the conversion for average carbon sequestration is estimated to be 0.97-ton CO2e/m<sup>3</sup> (ibid).

Therefore, the contribution to the total economic value from an increase of one unit for each of the unique variable of interest shows that the most significant effect occurs when there is a change in the price of carbon sequestration. Further crucial changes arise following variations in the price of timber and conversion factor. For example, considering the carbon price to be the same as the European carbon market price of 300 SEK/ ton CO2ev (table A1). The total economic values increase by 11% in the base scenario, by 18.5% for an average contribution of a one-unit increase in temperatures and by 12% for an average contribution of a one-unit increase in length vegetation period compared to their respective initial benefit values. Similarly, the value of carbon sequestration rises remarkably to almost double the initial carbon value in the three different scenarios with an increase of 85% extra benefit.

For the minimum price of timber to be 300 SEK m3/ha, the timber value reduces by about 17% of its initial value. Besides that, the total economic benefit changes downwards by 14.5% in the base scenario, by 13% with an increment of the marginal unit of temperature and by a reduction of 14 % with an increase of marginal unit of length of the vegetation period relative to their initial benefit values. Halving the value of the conversion factor to 0.48, the value of carbon sequestration is also halved to attain 50% of its primary economic value.

These results depend heavily on the underlying assumptions of prices of the forest ecosystem services of this study and the use of simple sigmoid growth function. Notably, this growth function ignores the composition of tree ages, the unchanged intrinsic growth rate, and the relative interconnectivity between volume growth and pressure intensity (e.g., Clark 2010). Using other growth functions, such as age-structured models could produce different predictions of forest growth since the Swedish forests are even-aged, and the age structure in a particular area has a tremendous impact on the forest growth. Also, more variables of interest need to be included in order to estimate such a function at the national scale. Despite that, the estimates of intrinsic forest growth rate for Sweden in this study are closer to the results found in other studies, such as Amuakwa-Mensah *et al.* (2018) and Eriksson & Vesterberg (2016).

Another critical assumption in this study is the conditional-response in forest management treatments if the introduction of compensation policy for carbon sequestration in Sweden is expected in comparison to some other countries (Gren & Aklilu 2016; Gren & Amuakwa-Mensah 2018). To simplify, timber, as well as carbon sequestration, have been associated with harvest. Of course, carbon sequestration is more complicated and depends on biomass growth, tree species, and soil conditions to mention a few. Therefore, introducing payments for ecosystem services such as carbon sequestration would likely result in changing management treatments and harvesting practices with the purpose to optimize the long term benefits of forest ecosystems of both timber and carbon sequestration (Gren & Aklilu 2016). Admittedly, these changes can be expected to increase the total economic benefit for the forests.

These present results illustrate a clear implication on the national forest policy. The policy aims to increase timber productivity equally the same as protecting biodiversity. With the use of the adjusted growth model, the results show that the value of timber remains constant with a oneunit increase of temperature and length vegetation period, whereas the value of carbon sequestration increases following the increases of the marginal increase in temperature and length of the vegetation period. The main explanation for these results to differ partly from the objectives of the policy is that the policy relies on the optimal management practices and the regression results have shown that scarification is the only management treatment which has significant effects on the forest growth. Furthermore, the increase in the value of carbon storage partly follows the long run harvesting decision under the present estimation method.

## 7 Conclusions

With the aim of this study, the findings illustrate that temperature and length of vegetation period are critical determinants of the economic value of carbon sequestration, which is one of the Swedish forest ecosystems studied. The use of the adjusted growth model could not verify the contribution of similar significant weather variability on the value of timber. It turns out that these variables of interest have long-term effects on boreal forest growth and therefore increase forest productivity through carbon sequestration. The value of timber and carbon sequestration from Swedish forests indicate that the average contribution from a one-unit increase in temperature amounts to SEK 1181/m<sup>3</sup> of the total economic benefits which correspond to 11% increase of the total initial value with an estimate of SEK 1065/m<sup>3</sup>. On the other hand, the contribution from a one-unit increase of the length of the vegetation period amounted SEK 1081/m<sup>3</sup>, which corresponds to an increase of 1.5% from the primary total benefit value in the base scenario.

The results show that the economic value of timber production in the entire country is equivalent to 23% of the gross domestic product in the year 2018. Unfortunately, the study could not compute the overall total economic value at the national level because of the nature of the weather conditions which vary across different regions in the country. The total economic value depends relatively to the assumptions on the prices and the value of the conversion factor.

These results are valid only for similar boreal conditions, but the methodological approach can be applied commonly to any other national scale where the average parameters' values of forest ecosystems are of paramount concern. Also, the positive of marginal effects on the adjusted growth rate imply that they are essential factors for optimal management activities relating to timber production and harvest practices. One area of interesting future research could be the optimization of management decisions that would allow long-term impacts on sustainable forest ecosystems. More explicitly, the use of quantitative interactions can help forests to provide a full range of potential economic benefits.

#### References

- Alexandersson, H., Ivarsson, K.-I. & Edquist, E. (2019-07-18). Stormen Gudrun, januaristormen 2005 | SMHI. SMHI.SE. Available at:
  - https://www.smhi.se/publikationer/januaristormen-2005-1.6399 [2019-08-29]
- Amuakwa-Mensah, F. (2019). Discussion with Amuaka-Mensah on extrapolating the total value of Carbon sequestration at a national level. [2019-08-26]
- Amuakwa-Mensah, F., Gren, I.-M. & Li, C.-Z. (2016). Fungi diversity in Swedish forests: impact of site quality on standing volume and variability in forest growth. p. 36
- Barbaroux, C. & Breda, N. (2002). Contrasting distribution and seasonal dynamics of carbohydrate reserves in stem wood of adult ring-porous sessile oak and diffuseporous beech trees. *Tree Physiology*, vol. 22 (17), pp. 1201–1210. DOI: https://doi.org/10.1093/treephys/22.17.1201
- Bellassen, V. & Luyssaert, S. (2014). Carbon sequestration: Managing forests in uncertain times. *Nature*, vol. 506 (7487), pp. 153–155. DOI: https://doi.org/10.1038/506153a
- Bergh, J., Freeman, M., Sigurdsson, B., Kellomäki, S., Laitinen, K., Niinistö, S., Peltola, H. & Linder, S. (2003). Modelling the short-term effects of climate change on the productivity of selected tree species in Nordic countries. *Forest Ecology and Management*, vol. 183 (1–3), pp. 327–340. DOI: https://doi.org/10.1016/S0378-1127(03)00117-8
- Bjørnstad, E. & Skonhoft, A.Wood Fuel or Carbon Sink? Aspects of Forestry in the Climate Question. p. 20
- Brainard, J., Bateman, I.J. & Lovett, A.A. (2009). The social value of carbon sequestered in Great Britain's woodlands. *Ecological Economics*, vol. 68 (4), pp. 1257–1267. DOI: https://doi.org/10.1016/j.ecolecon.2008.08.021
- Carlsson, M. (2012). Bioenergy from the Swedish Forest Sector. A Partial Equilibrium Analysis of Supply Costs and Implications for the Forest Product Markets. WORKING PAPER10/2012. Uppsala: SLU Sveriges lantbruksuniversitet, Institutionen för ekonom. Available at: https://pub.epsilon.slu.se/9117/1/carlsson m 121004.pdf
- Christiansen, L. & Fong Ekistrand, A. (2013). *Skogsstatistisk årsbok 2013 (Swedish Statistical Yearbook of Forestry)*. Jönköping: Swedish Forest Agency, Sweden. Available at: www.skogsstyrelsen.se/statistik
- Christiansen, L. & Fong Ekistrand, A. (2014). *Skogsstatistisk årsbok 2014 (Swedish Statistical Yearbook of Forestry)*. Jönköping: Swedish Forest Agency, Sweden. Available at: www.skogsstyrelsen.se/statistik
- Clark, C.W. (2010). *Mathematical bioeconomics: the mathematics of conservation*. 3rd ed. Hoboken, N.J: Wiley. (Pure and applied mathematics)
- Conley, D.J., Björck, S., Bonsdorff, E., Carstensen, J., Destouni, G., Gustafsson, B.G., Hietanen, S., Kortekaas, M., Kuosa, H., Markus Meier, H.E., Müller-Karulis, B., Nordberg, K., Norkko, A., Nürnberg, G., Pitkänen, H., Rabalais, N.N., Rosenberg, R., Savchuk, O.P., Slomp, C.P., Voss, M., Wulff, F. & Zillén, L. (2009). Hypoxia-Related Processes in the Baltic Sea. *Environmental Science & Technology*, vol. 43 (10), pp. 3412–3420. DOI: https://doi.org/10.1021/es802762a
- Coops, N.C., Waring, R.H. & Landsberg, J.J. (1998). Assessing forest productivity in Australia and New Zealand using a physiologically-based model driven with averaged monthly weather data and satellite-derived estimates of canopy photosynthetic capacity. *Forest Ecology and Management*, vol. 104 (1), pp. 113–127. DOI: https://doi.org/10.1016/S0378-1127(97)00248-X

- Daly, C., Bachelet, D., Lenihan, J.M., Neilson, R.P., Parton, W. & Ojima, D. (2000). DYNAMIC SIMULATION OF TREE–GRASS INTERACTIONS FOR GLOBAL CHANGE STUDIES. *Ecological Applications*, vol. 10 (2), pp. 449–469. DOI: https://doi.org/10.1890/1051-0761(2000)010[0449:DSOTGI]2.0.CO;2
- Eriksson, E. (2006). Thinning operations and their impact on biomass production in stands of Norway spruce and Scots pine. *Biomass and Bioenergy*, vol. 30 (10), pp. 848–854. DOI: https://doi.org/10.1016/j.biombioe.2006.04.001
- Eriksson, M., Hardell, L., Carlberg, M. & Åkerman, M. (2008). Pesticide exposure as risk factor for non-Hodgkin lymphoma including histopathological subgroup analysis. *International Journal of Cancer*, vol. 123 (7), pp. 1657–1663. DOI: https://doi.org/10.1002/ijc.23589
- Eriksson, M. & Vesterberg, A. (2016). *When Not in the Best of Worlds: Uncertainty and Forest Carbon Sequestration*. (ID 2766236). Rochester, NY: Social Science Research Network. Available at: https://papers.ssrn.com/abstract=2766236 [2019-03-28]
- Fekedulegn, D., Mac Siurtain, M. & Colbert, J. (1999). Parameter estimation of nonlinear growth models in forestry. *Silva Fennica*, vol. 33 (4). DOI: https://doi.org/10.14214/sf.653
- Fox, T.R., Lee Allen, H., Albaugh, T.J., Rubilar, R. & Carlson, C.A. (2007). Tree Nutrition and Forest Fertilization of Pine Plantations in the Southern United States. *Southern Journal of Applied Forestry*, vol. 31 (1), pp. 5–11. DOI: https://doi.org/10.1093/sjaf/31.1.5
- Gren, I.-M. & Aklilu, A.Z. (2016). Policy design for forest carbon sequestration: A review of the literature. *Forest Policy and Economics*, vol. 70, pp. 128–136. DOI: https://doi.org/10.1016/j.forpol.2016.06.008
- Gren, I.-M. & Amuakwa-Mensah, F. (2018). Estimating economic value of site quality for uncertain ecosystem service provision in Swedish forests. *International Journal of Biodiversity Science, Ecosystem Services & Management*, vol. 14 (1), pp. 117–126. DOI: https://doi.org/10.1080/21513732.2018.1472143
- Gren, I.-M., Häggmark-Svensson, T., Andersson, H., Jansson, G. & Jägerbrand, A. (2016).
  Using traffic data to estimate wildlife populations. *Journal of Bioeconomics*, vol. 18 (1), pp. 17–31. DOI: https://doi.org/10.1007/s10818-015-9209-0
- Haanpää, S., Lehtonen, S., Peltonen, L. & Talockaite, E. (2005). Impacts of winter storm Gudrun of 7th –9th January 2005 and measures taken in Baltic Sea Region. p. 43. Available at: http://discomap.eea.europa.eu/map/Data/Milieu/OURCOAST 110 Baltic/OURCOA
  - ST\_110\_Baltic\_Doc1\_ImpactGudrunStorm.pdf
- Hastie, T. & Tibshirani, R. (1986). Generalized Additive Models. *Statistical Science*, vol. 1 (3), pp. 297–310. DOI: https://doi.org/10.1214/ss/1177013604
- Helin, T., Sokka, L., Soimakallio, S., Pingoud, K. & Pajula, T. (2013). Approaches for inclusion of forest carbon cycle in life cycle assessment - a review. *GCB Bioenergy*, vol. 5 (5), pp. 475–486. DOI: https://doi.org/10.1111/gcbb.12016
- Högberg, P., Näsholm, T., Franklin, O. & Högberg, M.N. (2017). Tamm Review: On the nature of the nitrogen limitation to plant growth in Fennoscandian boreal forests. *Forest Ecology and Management*, vol. 403, pp. 161–185. DOI: https://doi.org/10.1016/j.foreco.2017.04.045
- Huuskonen, S. & Hynynen, J. (2006). Timing and intensity of precommercial thinning and their effects on the first commercial thinning in Scots pine stands. *Silva Fennica*, vol. 40 (4). DOI: https://doi.org/10.14214/sf.320

- Jacobson, S. & Pettersson, F. (2010). An assessment of different fertilization regimes in three boreal coniferous stands. *Silva Fennica*, vol. 44 (5). DOI: https://doi.org/10.14214/sf.123
- Janssens, I.A., Freibauer, A., Schlamadinger, B., Ceulemans, R., Ciais, P., Dolman, A.J., Heimann, M., Nabuurs, G.-J., Smith, P., Valentini, R. & Schulze, E.-D. (2005). The carbon budget of terrestrial ecosystems at country-scale – a European case study. *Biogeosciences*, vol. 2 (1), pp. 15–26. DOI: https://doi.org/10.5194/bg-2-15-2005
- Johansson, K., Nilsson, U. & Allen, H.L. (2006). Interactions between soil scarification and Norway spruce seedling types. *New Forests*, vol. 33 (1), pp. 13–27. DOI: https://doi.org/10.1007/s11056-006-9010-y
- Johansson, K., Nilsson, U. & Orlander, G. (2013a). A comparison of long-term effects of scarification methods on the establishment of Norway spruce. *Forestry*, vol. 86 (1), pp. 91–98. DOI: https://doi.org/10.1093/forestry/cps062
- Johansson, K., Ring, E. & Högbom, L. (2013b). Effects of pre-harvest fertilization and subsequent soil scarification on the growth of planted Pinus sylvestris seedlings and ground vegetation after clear-felling. *Silva Fennica*, vol. 47 (4). DOI: https://doi.org/10.14214/sf.1016
- Johansson, M. (1994). The influence of soil scarification on the turn-over rate of slash needles and nutrient release. *Scandinavian Journal of Forest Research*, vol. 9 (1–4), pp. 170–179. DOI: https://doi.org/10.1080/02827589409382828
- Juodvalkis, A., Kairiukstis, L. & Vasiliauskas, R. (2005). Effects of thinning on growth of six tree species in north-temperate forests of Lithuania. *European Journal of Forest Research*, vol. 124 (3), pp. 187–192. DOI: https://doi.org/10.1007/s10342-005-0070-x
- Kellomäki et al., S. (2005). Adaptation of forest ecosystems, forests and forestry to climate change. Helsinki: Finnish Environment Institute. Available at: https://helda.helsinki.fi/bitstream/handle/10138/41042/SYKEmo\_334.pdf?sequence= 1&isAllowed=y
- Kellomäki, S. & Väisänen, H. (1997). Modelling the dynamics of the forest ecosystem for climate change studies in the boreal conditions. *Ecological Modelling*, vol. 97 (1–2), pp. 121–140. DOI: https://doi.org/10.1016/S0304-3800(96)00081-6
- Landsberg, J.J. & Waring, R.H. (2014). Forests in our changing world: new principles for conservation and management. Washington [D.C.]: Island Press.
- Larson, P.R., Kretschmann, D.E., Clark, A.I. & Isebrands, J.G. (2001). Formation and properties of juvenile wood in southern pines: a synopsis. (FPL-GTR-129). Madison, WI: U.S. Department of Agriculture, Forest Service, Forest Products Laboratory. DOI: https://doi.org/10.2737/FPL-GTR-129
- Lindkvist, A., Kardell, Ö. & Nordlund, C. (2011). Intensive Forestry as Progress or Decay? An Analysis of the Debate about Forest Fertilization in Sweden, 1960–2010. *Forests*, vol. 2 (1), pp. 112–146. DOI: https://doi.org/10.3390/f2010112
- (LULUCF) Utsläpp och upptag av växthusgaser från markanvändning. (2018). Naturvårdsverket. [text]. Available at: http://www.naturvardsverket.se/Sa-marmiljon/Statistik-A-O/Vaxthusgaser-utslapp-och-upptag-fran-markanvandning/ [2019-08-29]
- Lundmark, T., Bergh, J., Hofer, P., Lundström, A., Nordin, A., Poudel, B., Sathre, R., Taverna, R. & Werner, F. (2014). Potential Roles of Swedish Forestry in the Context of Climate Change Mitigation. *Forests*, vol. 5 (4), pp. 557–578. DOI: https://doi.org/10.3390/f5040557
- Luyssaert, S., Inglima, I., Jung, M., Richardson, A.D., Reichstein, M., Papale, D., Piao, S.L., Schulze, E.-D., Wingate, L., Matteucci, G., Aragao, L., Aubinet, M., Beer, C., Bernhofer, C., Black, K.G., Bonal, D., Bonnefond, J.-M., Chambers, J., Ciais, P.,

Cook, B., Davis, K.J., Dolman, A.J., Gielen, B., Goulden, M., Grace, J., Granier, A., Grelle, A., Griffis, T., Grünwald, T., Guidolotti, G., Hanson, P.J., Harding, R., Hollinger, D.Y., Hutyra, L.R., Kolari, P., Kruijt, B., Kutsch, W., Lagergren, F., Laurila, T., Law, B.E., Le Maire, G., Lindroth, A., Loustau, D., Malhi, Y., Mateus, J., Migliavacca, M., Misson, L., Montagnani, L., Moncrieff, J., Moors, E., Munger, J.W., Nikinmaa, E., Ollinger, S.V., Pita, G., Rebmann, C., Roupsard, O., Saigusa, N., Sanz, M.J., Seufert, G., Sierra, C., Smith, M.-L., Tang, J., Valentini, R., Vesala, T. & Janssens, I.A. (2007). CO <sub>2</sub> balance of boreal, temperate, and tropical forests derived from a global database. *Global Change Biology*, vol. 13 (12), pp. 2509–2537. DOI: https://doi.org/10.1111/j.1365-2486.2007.01439.x

- MacDonald, P. (2016). Sandbag. *Sandbag.org.uk*. (smarter climate policy). Available at: https://sandbag.org.uk/2016/09/02/eu-carbon-price-falls-below-e4-2/
- Makinen, H. (2004). Thinning intensity and growth of Norway spruce stands in Finland. *Forestry*, vol. 77 (4), pp. 349–364. DOI: https://doi.org/10.1093/forestry/77.4.349
- Mäkinen, H. & Isomäki, A. (2004). Thinning intensity and growth of Scots pine stands in Finland. *Forest Ecology and Management*, vol. 201 (2–3), pp. 311–325. DOI: https://doi.org/10.1016/j.foreco.2004.07.016
- Mattsson, S. & Bergsten, U.Pinus contorta growth in northern Sweden as affected by soil scarification. p. 16
- McDermott, S.M., Howarth, R.B. & Lutz, D.A. (2015). Biomass Energy and Climate Neutrality: The Case of the Northern Forest. *Land Economics*, vol. 91 (2), pp. 197– 210. DOI: https://doi.org/10.3368/le.91.2.197
- Munich Re 2002*Munich Re Group Annual Report 2002.* (302–03661). Munichen, Germany. Available at: https://www.munichre.com/site/corporate/get/params\_E-369054748/1390154/302-03661\_en.pdf
- Nguimkeu, P. (2014). A simple selection test between the Gompertz and Logistic growth models. *Technological Forecasting and Social Change*, vol. 88, pp. 98–105. DOI: https://doi.org/10.1016/j.techfore.2014.06.017
- Nguimkeu, P.E. & Rekkas, M. (2011). Third-order inference for autocorrelation in nonlinear regression models. *Journal of Statistical Planning and Inference*, vol. 141 (11), pp. 3413–3425. DOI: https://doi.org/10.1016/j.jspi.2011.04.019
- Nilsson, C., Stjernquist, I., Bärring, L., Schlyter, P., Jönsson, A.M. & Samuelsson, H. (2004). Recorded storm damage in Swedish forests 1901–2000. Forest Ecology and Management, vol. 199 (1), pp. 165–173. DOI: https://doi.org/10.1016/j.foreco.2004.07.031
- Nilsson, P., Cory, N. & Dahlgren, J. (2011). SKOGSDATA 2011. Aktuella uppgifter om de svenska skogarna från Riksskogstaxeringen. Tema: Fält- och bottenskiktsvegetation i Sveriges skogar. Umea: SLUInstitutionen för skoglig resurshushållning. Available at: https://pub.epsilon.slu.se/8333/1/Skogsdata2011\_webb.pdf
- Nilsson, P., Cory, N., Fridman, J. & Wulff, S. (2018). SKOGSDATA 2018. Aktuella uppgifter om de svenska skogarna från Riksskogstaxeringen. Tema: Riksskogstaxeringens kvalitetssäkringsarbete. Umea: SLU Institutionen för skoglig resurshushållning. Available at:

https://www.slu.se/globalassets/ew/org/centrb/rt/dokument/skogsdata/skogsdata\_2018 \_webb.pdf

Nilsson, P., Cory, N. & Stendahl, J. (2017). *Skogsdata 2017 aktuella uppgifter om de svenska skogarna från Riksskogstaxeringen. Tema: Skogsmarkens kolförråd.* Umea: SLU. Institutionen för skoglig resurshushållning. Available at: https://www.slu.se/globalassets/ew/org/centrb/rt/dokument/skogsdata/skogsdata 2017

https://www.slu.se/globalassets/ew/org/centrb/rt/dokument/skogsdata/skogsdata\_2017 .pdf

- Nilsson, U. (2010). Thinning of Scots pine and Norway spruce monocultures in Sweden effects of different thinning programmes on stand level gross- and net stem volume production. Umea: Swedish University of Agricultural Sciences, Faculty of Forest Sciences. Available at: http://pub.epsilon.slu.se/4655/1/SFS219.pdf [2019-05-09]
- Nilsson, U. & Allen, H.L. (2003). Short- and long-term effects of site preparation, fertilization and vegetation control on growth and stand development of planted loblolly pine. *Forest Ecology and Management*, vol. 175 (1–3), pp. 367–377. DOI: https://doi.org/10.1016/S0378-1127(02)00140-8
- Nilsson, U. & örlander, G. (1999). Water Uptake by Planted Picea abies in Relation to Competing Field Vegetation and Seedling Rooting Depth on Two Grass-dominated Sites in Southern Sweden. *Scandinavian Journal of Forest Research*, vol. 14 (4), pp. 312–319. DOI: https://doi.org/10.1080/02827589950152629
- Nolander, C. (2018). Spatial and Economic Values of Ecosystem Services The Case of Sweden's Forests. Luleå University of Technology. Available at: http://www.divaportal.org/smash/record.jsf?pid=diva2%3A1190017&dswid=-1686
- Nolander, C.Spatial and Economic Values of Ecosystem Services. p. 98
- Noormets, A., Epron, D., Domec, J.C., McNulty, S.G., Fox, T., Sun, G. & King, J.S. (2015). Effects of forest management on productivity and carbon sequestration: A review and hypothesis. *Forest Ecology and Management*, vol. 355, pp. 124–140. DOI: https://doi.org/10.1016/j.foreco.2015.05.019
- Nordborg, F., Nilsson, U., Gemmel, P. & Örlander, G. (2006). Carbon and nitrogen stocks in soil, trees and field vegetation in conifer plantations 10 years after deep soil cultivation and patch scarification. *Scandinavian Journal of Forest Research*, vol. 21 (5), pp. 356–363. DOI: https://doi.org/10.1080/02827580600976615
- Nordborg, F., Nilsson, U. & Örlander, G. (2003). Effects of different soil treatments on growth and net nitrogen uptake of newly planted Picea abies (L.) Karst. seedlings. *Forest Ecology and Management*, vol. 180 (1–3), pp. 571–582. DOI: https://doi.org/10.1016/S0378-1127(02)00650-3
- Nordlander, G., Hellqvist, C., Johansson, K. & Nordenhem, H. (2011). Regeneration of European boreal forests: Effectiveness of measures against seedling mortality caused by the pine weevil Hylobius abietis. *Forest Ecology and Management*, vol. 262 (12), pp. 2354–2363. DOI: https://doi.org/10.1016/j.foreco.2011.08.033
- Örlander, G., Gemmel, P. & Hunt, J. (1990). *Site preparation: a Swedish overview*. Victoria, B.C: Forestry Canada, Pacific Forestry Centre. (FRDA report; 105)
- Persson, P. (1975). Windthrow in forests its causes and the effect of forestry measures [Sweden, Scots pine, Norway spruce]. (Thesis). SLU University Library, Alnarp. Available at: Journal of The Forestry Academy, 42. ISSN: 0585-3303 [2019-07-01]
- Phillips, P.C.B. & Hansen, B.E. (1990). Statistical Inference in Instrumental Variables Regression with I(1) Processes. *The Review of Economic Studies*, vol. 57 (1), p. 99. DOI: https://doi.org/10.2307/2297545
- *Riksbank. Statistics* (2019). Stockholm. Available at: https://www.riksbank.se/engb/statistics/search-interest--exchange-rates/annual-average-exchange-rates/
- Ring, E. (2004). Experimental N fertilization of Scots pine: effects on soil-solution chemistry 8 years after final felling. *Forest Ecology and Management*, vol. 188 (1–3), pp. 91–99. DOI: https://doi.org/10.1016/j.foreco.2003.07.020
- Saarsalmi, A. & Tamminen, P. (2005). Boron, phosphorus and nitrogen fertilization in Norway spruce stands suffering from growth disturbances. *Silva Fennica*, vol. 39 (3). DOI: https://doi.org/10.14214/sf.373

- Saarsalmi, A., Tamminen, P. & Kukkola, M. (2014). Effects of long-term fertilisation on soil properties in Scots pine and Norway spruce stands. *Silva Fennica*, vol. 48 (1). DOI: https://doi.org/10.14214/sf.989
- Savill, P.S. (ed.) (1997). *Plantation silviculture in Europe*. Oxford [England]; New York: Oxford University Press.
- Scholze, M., Knorr, W., Arnell, N.W. & Prentice, I.C. (2006). A climate-change risk analysis for world ecosystems. *Proceedings of the National Academy of Sciences*, vol. 103 (35), pp. 13116–13120. DOI: https://doi.org/10.1073/pnas.0601816103
- Sedjo, R. & Tian, X. (2012). Does Wood Bioenergy Increase Carbon Stocks in Forests? Journal of Forestry, vol. 110 (6), pp. 304–311. DOI: https://doi.org/10.5849/jof.11-073
- SEPA (2012). Sweden's Environmental Objectives An Introduction. Stockholm. Available at: http://www.swedishepa.se/Documents/publikationer6400/978-91-620-8820-0.pdf?pid=23427
- SFA (2015). Forests and Forestry in Sweden. Stockholm. Available at: https://www.skogsstyrelsen.se/globalassets/in-english/forests-and-forestry-insweden\_2015.pdf
- smhi.se (2018). https://www.smhi.se/kunskapsbanken/meteorologi/hur-mats-lufttemperatur-1.3839. Available at: https://www.smhi.se/kunskapsbanken/meteorologi/hur-matslufttemperatur-1.3839
- smhi.se (2019-08-20). Climate indicators length of vegetation period | SMHI. Swedish Meteological and Hydrological Institute. Available at: https://www.smhi.se/en/climate/climate-indicators/climate-indicators-length-ofvegetation-period-1.91482 [2019-08-29]
- smhi.se *Meteorologiska observationer Vindhastighet timvärde*. Available at: https://www.smhi.se/data/meteorologi/ladda-ner-meteorologiskaobservationer/#param=wind,stations=all [2019-10-14]
- Sohngen, B. & Mendelsohn, R. (2003). An Optimal Control Model of Forest Carbon Sequestration. American Journal of Agricultural Economics, vol. 85 (2), pp. 448–457. DOI: https://doi.org/10.1111/1467-8276.00133
- Tapio (2013). *Metsänhoidon suositukset*. Helsinki, Finland: Metsäkustannus Oy. Available at: http://tapio.fi/wp-

content/uploads/2015/06/Metsanhoidon\_suositukset\_ver3\_netti\_1709141.pdf

- Thiffault, N., Titus, B.D. & Munson, A.D. (2005). Silvicultural options to promote seedling establishment on Kalmia – Vaccinium -dominated sites. Scandinavian Journal of Forest Research, vol. 20 (2), pp. 110–121. DOI: https://doi.org/10.1080/02827580510008356
- Tol, R.S.J. (2013). Targets for global climate policy: An overview. Journal of Economic Dynamics and Control, vol. 37 (5), pp. 911–928. DOI: https://doi.org/10.1016/j.jedc.2013.01.001
- Vass, M.M. & Elofsson, K. (2016). Is forest carbon sequestration at the expense of bioenergy and forest products cost-efficient in EU climate policy to 2050? *Journal of Forest Economics*, vol. 24, pp. 82–105. DOI: https://doi.org/10.1016/j.jfe.2016.04.002
- Wallentin, C. (2007). *Thinning of Norway spruce*. Swedish University of Agricultural Sciences.
- Wang, H. (2006). Extending the Linear Model with R: Generalized Linear, Mixed Effects and Nonparametric Regression Models Edited by Faraway J. J. *Biometrics*, vol. 62 (4), pp. 1278–1278. DOI: https://doi.org/10.1111/j.1541-0420.2006.00596\_12.x

- Wang, Q. & Wu, N. (2012). Long-Run Covariance and its Applications in Cointegration Regression. *The Stata Journal: Promoting communications on statistics and Stata*, vol. 12 (3), pp. 515–542. DOI: https://doi.org/10.1177/1536867X1201200312
- Way, D.A. & Oren, R. (2010). Differential responses to changes in growth temperature between trees from different functional groups and biomes: a review and synthesis of data. *Tree Physiology*, vol. 30 (6), pp. 669–688. DOI: https://doi.org/10.1093/treephys/tpq015
- WITS (2017). World | Wood | Exports and Imports | by all Countries | 2017 | WITS | Data. World Integrated Trade Solution. Available at: https://wits.worldbank.org/CountryProfile/en/Country/WLD/Year/2017/TradeFlow/E XPIMP/Partner/all/Product/44-49\_Wood [2019-08-29]
- Wood, S.N. (2017). *Generalized Additive Models : An Introduction with R, Second Edition*. Chapman and Hall/CRC. DOI: https://doi.org/10.1201/9781315370279
- World Bank (2019-08-31). GDP (current US\$). Sweden World Bank national accounts data, and OECD National Accounts data files. [worldbank.org]. Available at: https://data.worldbank.org/indicator/NY.GDP.MKTP.CD?locations=SE&view=chart
- Xu, C.-Y. & Griffin, K.L. (2006). Seasonal variation in the temperature response of leaf respiration in Quercus rubra: foliage respiration and leaf properties. *Functional Ecology*, vol. 20 (5), pp. 778–789. DOI: https://doi.org/10.1111/j.1365-2435.2006.01161.x

## Acknowledgment

This thesis as the final part of my study has been produced during my scholarship period at Swedish University of Agricultural Sciences (SLU), thanks to the Swedish Institute Scholarship (SI).

I would like to thank my lovely family and friends for your support, critical reading of this thesis.

A special thank goes to my supervisor Franklin Amuakwa Mensah who has guided me, and for useful discussions during my master's thesis.

#### Appendix

#### Figure A1: Forest zones and counties in Sweden



Source: Swedish Forest Data, 2014 report. SLU, Forest Resources Management Department NB: Norra Norrland and Södra Norrland are North Norrland and South Norrland, respectively

| Table A1: Calculated total economic value and average contribution of a one-unit increase of         |
|--|
| temperature and length vegetation period to the total economic value of timber and carbon            |
| sequestration for changes in price of timber $(P^T)$ , price of carbon sequestration $(P^{CO2})$ and |
| conversion factor ( <i>conv</i> ) in SEK/m <sup>3</sup> /ha  |

| Variables/ Economic value Contribution of a one unit increment |               |             |                          |  |  |  |
|--|---------------|-------------|--------------------------|--|--|--|
| Values   | Base scenario | temperature | length vegetation period |  |  |  |
|  |               |             |                          |  |  |  |
| $P^{T} = 300$  |               |             |                          |  |  |  |
| Timber:  | 769.35        | 769.35      | 769.35                   |  |  |  |
| Carbon Sequestration:  | 141.55        | 258.01      | 157.76                   |  |  |  |
| Total:   | 910.90        | 1027.36     | 927.12                   |  |  |  |

| $P^{CO2} = 300$       |         |         |         |  |
|-----------------------|---------|---------|---------|--|
| Timber:               | 923.22  | 923.22  | 923.22  |  |
| Carbon Sequestration: | 262.13  | 477.80  | 292.15  |  |
| Total:                | 1185.35 | 1401.02 | 1215.38 |  |
| <i>Conv</i> =0.485    |         |         |         |  |
| Timber:               | 923.22  | 923.22  | 923.22  |  |
| Carbon Sequestration: | 70.77   | 129.01  | 78.88   |  |
| Total:                | 993.99  | 1052.23 | 1002.10 |  |

Table A2: The total economic benefit of timber production due to an average contribution of a one-unit increase of temperature and the length of the vegetation period in SEK.

 $P^{T}$ = 360 Timber: 3.00 10<sup>9</sup>\* 360= 108.00 10<sup>10</sup> SEK

P<sup>T</sup>=300 Timber: 3.00 10<sup>6</sup> \*300= 90.00 10<sup>10</sup> SEK

Table A3: The comparison of the total economic value of the timber production and the gross domestic product (GDP) in 2018.

 $P^{T}$ = 360 Timber: 3.00 10<sup>9</sup>\* 360= 108.00 10<sup>10</sup> SEK

GDP (2018): 551.032 10<sup>9</sup>\*8.5315= 470.1129508 10<sup>10</sup> SEK (TOTAL TIMBER VALUE /GDP (2018))\*100= 22.97% of the GDP (2018).

 $P^{T}$ = 300 Timber: 3.00 10<sup>9</sup>\* 300= 90.00 10<sup>10</sup> SEK

GDP (2018): 551.032 10<sup>9</sup>\*8.5315= 470.1129508 10<sup>10</sup> SEK (TOTAL TIMBER VALUE /GDP (2018))\*100= 19% of the GDP (2018).